Author’s response to reviews

Title: Risk of Zika virus transmission in the Euro-Mediterranean area and the added value of building preparedness to arboviral threats from a One Health perspective

Authors:
Camille Escadafal (camille.escadafal@pasteur.fr)
Lobna Gaayeb (lobna.gaayeb@pasteur.fr)
Flavia Riccardo (flavia.riccardo@iss.it)
Elisa Pérez-Ramírez (elisaperezramirez@gmail.com)
Marie Picard (marie.picard@ird.fr)
Maria Grazia Dente (mariagrazia.dente@iss.it)
Jean-Calude Manuguerra (jean-claude.manuguerra@pasteur.fr)
Miguel Ángel Jiménez-Clavero (majimenez@inia.es)
Silvia Declich (silvia.decllich@iss.it)
Kathleen Victoir (kathleen.victoir@pasteur.fr)
Vincent Robert (vincent.robert@ird.fr)

Version: 3 Date: 04 Nov 2016

Author’s response to reviews:

Dear editor and reviewers, thank you for your comments.

Please find hereafter our responses.

1) Please state whether Figure 1 is your own or if it has been taken from another source.

Figure 1 is our own. A specific sentence has been added to the figure legend, at the end of the manuscript: "Figure 1 is produced by the authors of the present manuscript."
2) Please ensure that there is a section present in your Declarations titled "Acknowledgements".

The section has been included in the new version of the manuscript.

3) At this stage, please upload your revised manuscript as a single, clean copy with no tracked changes as these are no longer required at this stage of the editorial process.

A "clean" copy has been uploaded